Author’s response to reviews

Title: Nitrates and Bone Turnover [NABT]: trial to select the best nitrate preparation.

Authors:

Celeste J Hamilton (celeste.hamilton@wchospital.ca)
Steven R Cummings (scummings@sfcc-cpmc.net)
Sophie A Jamal (sophie.jamal@utoronto.ca)

Version: 2 Date: 17 September 2012

Author’s response to reviews:

Dear Editor,

This revision is to include our proof of ethics approval and funding for our randomized controlled trial. We have included 2 additional documents.

Regards,

Dr. Sophie A. Jamal